Author's response to reviews

Title: Biofilm Formation by Clinical Isolates and the Implications in Chronic Infections

Authors:

Carlos J Sanchez Jr. (carlos.j.sanchez3@us.army.mil)
Katrin Mende (katrin.mende@us.army.mil)
Miriam L. Beckius (miriam.l.beckius@us.army.mil)
Kevin S. Akers (kevin.akers@us.army.mil)
Desiree R. Romano (desiree.r.romano.ctr@us.army.mil)
Joseph C. Wenke (joseph.wenke@us.army.mil)
Clinton K. Murray (clinton.murray@us.army.mil)

Version: 2 Date: 7 September 2012

Author's response to reviews:

BMC Editorial Staff,

The clinical isolates that were used in this study were samples that were collected from a clinical infectious disease strain repository. Bacterial strains, were collected from patients as part of the standard care and part of the infection control policy within the facility not related to research. Importantly, the strains used herein were not directly sampled from the patients at any time during treatment. Importantly, all attempts were made to remove any identifying information. The methods section (bacterial isolates and growth conditions) has been reworded to better reflect this.

Sincerely,

Carlos J Sanchez